OROS-methylphenidate to reduce ADHD symptoms in male prisoners aged 16–25 years: a RCT

Philip Asherson,^{1,2,3*} Lena Johansson,¹
Rachel Holland,⁴ Megan Bedding,¹
Andrew Forrester,^{5,6} Laura Giannulli,⁷ Ylva Ginsberg,⁸
Sheila Howitt,^{7,9} Imogen Kretzschmar,¹
Stephen Lawrie,⁷ Craig Marsh,¹⁰ Caroline Kelly,⁷
Megan Mansfield,¹ Clare McCafferty,⁹ Khuram Khan,⁹
Ulrich Müller-Sedgwick,¹¹ John Strang,¹²
Grace Williamson,¹ Lauren Wilson,¹ Susan Young,¹³
Sabine Landau⁴ and Lindsay Thomson^{7,9}

Declared competing interests of authors: Philip Asherson reports grants and personal fees from Janssen-Cilag Ltd (High Wycombe, UK), Medice (Iserlohn, Germany), Shire/Takeda Pharmaceutical Company Ltd (Tokyo, Japan) and Flynn Pharma Ltd (Stevenage, UK); non-financial support and grants from QbTech AB (Stockholm, Sweden); personal fees from Novartis International AG (Basel, Switzerland) and Eli Lilly and Company (Indianapolis, IN, USA); and grants from Vifor Pharma Group (Villars-sur-Glâne, Switzerland), GW Pharmaceuticals (Cambridge, UK) and QbTech AB, outside the submitted work. Ylva Ginsberg has received royalties, speaker fees, reimbursement for

¹Social, Genetic and Developmental Psychiatry Centre, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

²South London and Maudsley NHS Foundation Trust, London, UK

³Oxleas NHS Foundation Trust, Dartford, UK

⁴Department of Biostatistics and Health Informatics, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

⁵Greater Manchester Mental Health NHS Foundation Trust, Manchester, UK

⁶Offender Health Research Network, University of Manchester, Manchester, UK

⁷Division of Psychiatry, University of Edinburgh, Edinburgh, UK

⁸Department of Clinical Neuroscience, Centre for Psychiatry Research, Karolinska Institute, Stockholm, Sweden

⁹The Board, The State Hospital, Carstairs, UK

¹⁰NHS Forth Valley, Stirling, UK

¹¹Barnet, Enfield and Haringey Mental Health NHS Trust, London, UK

¹²Addiction Department, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

¹³Psychology Services Ltd, Croydon, UK

^{*}Corresponding author philip.asherson@kcl.ac.uk

travel costs and/or collaborated in research with Shire, Medscape [www.medscape.com/ (accessed 5 January 2021)] and Studentlitteratur AB (Lund, Sweden). Susan Young is director of Psychology Innovations Ltd (London, UK), which has received fees from Shire/Takeda, and has received fees for training in attention deficit hyperactivity disorder (ADHD) assessment tools and psychological interventions. She is a consultant at the Cognitive Centre of Canada and has received fees from the sale of psychological treatment programmes; she is also president of the UK ADHD Partnership, which received unrestricted educational grants from Shire/Takeda. Stephen Lawrie has received personal fees and research income from Janssen-Cilag, as well as personal fees from Sunovion Pharmaceuticals Inc. (Marlborough, MA, USA), in connection with work on schizophrenia (i.e. outside the submitted work). John Strang conducted research studies with the study medication being provided by the relevant pharmaceutical companies, and King's College London has received payment of consultancy fees or honoraria for such work. However, this work does not apply to the study of ADHD nor to the type of treatment (methylphenidate). Sabine Landau is supported by the NIHR Biomedical Research Centre at South London and Maudsley NHS Foundation Trust and King's College London, and the NIHR Applied Research Collaboration South London at King's College Hospital NHS Foundation Trust. Ulrich Müller-Sedgwick received honoraria for being on an advisory board and/or speaking/delivering training for the UK Adult ADHD Network, British Association for Psychopharmacology, Takeda, Sosei Heptares (London, UK), Eli Lilly and Company, and Flynn/Medice.

Published June 2022 DOI: 10.3310/THEI8200

Plain English summary

OROS-methylphenidate for ADHD symptoms in male prisoners

Efficacy and Mechanism Evaluation 2022; Vol. 9: No. 6

DOI: 10.3310/THEI8200

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

A round 20–30% of prisoners have attention deficit hyperactivity disorder (ADHD). Stimulants such as methylphenidate reduce symptoms of ADHD; however, it is not clear whether or not prisoners with ADHD show the same beneficial effects. Differences may arise because problems with poor attention, restlessness or impulsive behaviour might be better explained by other mental health conditions or drug use, or because comorbid disorders might interfere with their response to methylphenidate.

This study investigates the effects of methylphenidate among young male offenders with undiagnosed or untreated ADHD. Participants were randomised to either osmotic release oral system (OROS) methylphenidate (a sustained release form of methylphenidate) or placebo during an 8-week trial. Participants and all investigators were blinded to the treatment given. Symptoms of ADHD, as well as 13 other measures, including emotional dysregulation, general mental health symptoms and prison reports of behaviour, were measured throughout the trial.

The trial found no evidence of a benefit of OROS-methylphenidate, compared with placebo, on any of the outcomes considered. This is perhaps surprising because the use of methylphenidate in the treatment of ADHD is well established. The reasons for this lack of evidence will require further research, but could be due to participants not taking sufficient medication, or the high rate of co-occurring mental health, social and drug use problems. It is also feasible that the prison environment influences the findings if prisoners feel that they need to please prison or health-care staff to shorten their sentences or gain privileges. The trial findings do not support the routine treatment with OROS-methylphenidate of ADHD in prisoners.

Efficacy and Mechanism Evaluation

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

Efficacy and Mechanism Evaluation (EME) was launched in 2014 and is indexed by Europe PMC, DOAJ, Ulrichsweb and NCBI Bookshelf.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

EME programme

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme supports translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in man and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health and Care Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

This report

The research reported in this issue of the journal was funded by the EME programme as project number 14/23/17. The contractual start date was in August 2016. The final report began editorial review in August 2020 and was accepted for publication in December 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the EME programme or the Department of Health and Social Care.

Copyright © 2022 Asherson *et al.* This work was produced by Asherson *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk